Author’s response to reviews

Title: Ultrasound Assessment of Tensile Stress in Carotid Arteries of Healthy Human Subjects with Varying Age

Authors:

Xianghong Luo (lxh_20050703@sina.com)

Rong Wu (wurong7111@sina.com)

Lianfang Du (lf_du99@163.com)

Zhaojun Li (lzj_197506@126.com)

Version: 2 Date: 06 Nov 2019

Author’s response to reviews:

Thank you for your letter concerning our manuscript. We have read your comments carefully.

Answers to Reviewer


2. Availability of data and materials: The datasets are analyzed and are not publicly available, but are available from the corresponding author on reasonable request.